Author's response to reviews

Title: The SMILE study: A Study of Medical Information and Lifestyles in Eindhoven. The rationale and contents of a large prospective dynamic cohort study

Authors:

Marjan van den Akker (Marjan.vandenAkker@hag.unimaas.nl)
Mark G Spigt (M.Spijt@hag.unimaas.nl)
Lore De Raeve (Lore.deRaeve@epid.unimaas.nl)
Ben van Steenkiste (Ben.vanSteenkiste@hag.unimaas.nl)
Job FM Metsemakers (Job.Metsemakers@hag.unimaas.nl)
Ernst J van Voorst (E.vanvoorst@sge.nl)
Hein de Vries (Hein.deVries@gvo.unimaas.nl)

Version: 2 Date: 7 December 2007

Author's response to reviews:

Dear editor,

We would like to thank you for the opportunity to send you the revised version of our manuscript "The SMILE study: A Study of Medical Information and Lifestyles in Eindhoven. The rationale and contents of a large prospective dynamic cohort study."

As you suggested in your email we have removed the response rate and the baseline data. Furthermore, we have provided more technical details regarding data collection and data management (pages 7 and 8).

Finally, we have checked for language errors.

We sincerely hope that you will now be able to accept our manuscript for publication in BMC Public Health.

On behalf of all authors,
Marjan van den Akker